Author's response to reviews

Title: Association of angiotensin-converting enzyme inhibitor therapy and comorbidity in diabetes: results from the Vermont Diabetes Information System

Authors:

Maria E. Ramos-Nino (Maria.Ramos@uvm.edu)
Charles D. MacLean (charles.maclean@vtmednet.org)
Benjamin Littenberg (benjamin.littenberg@vtmednet.org)

Version: 2 Date: 15 September 2008

Author's response to reviews:

BMC-series Journals
BioMed Central
BMC Endocrine Disorders September 15, 2008

Dear Board:

The authors would appreciate the review of the manuscript by Ramos-Nino, MacLean and Littenberg entitled “Association of angiotensin-converting enzyme inhibitor therapy and comorbidity in diabetes: results from the Vermont Diabetes Information System”. This article is the product of a cross-sectional study of patients with diabetes in Northern New England. The data presented here suggests a potential protective effect of ACE inhibitors for cancer and peptic ulcer disease in these patients.

This article was originally submitted to BMC Medicine where they suggested that your subject-specific journal would be more appropriate for this article.

On behalf of my co-authors, thank you for your consideration.

Maria E Ramos-Nino, PhD
Research Assistant Professor
University of Vermont
Department of Pathology
Maria.Ramos@uvm.edu